Author's response to reviews

**Title:** Molecular characterization of EGFR, PDGFRα and VEGFR-2 in cervical adenosquamous carcinoma

**Authors:**

Adhemar Longatto-Filho (longatto@ecsaude.uminho.pt)  
Celine Pinheiro (cpinheiro@ecsaude.uminho.pt)  
Olga Martinho (olgamartinho@ecsaude.uminho.pt)  
Marise AR Moreira (mmoreira@yahoo.com)  
Luiz FJ Ribeiro (lfjribeiro@yahoo.com)  
Geraldo S Queiroz (queirozg@gmail.com)  
Fernando C Schmitt (fschmitt@ipatimup.pt)  
Fátima Baltazar (fbaltazar@ecsaude.uminho.pt)  
Rui M Reis (rreis@ecsaude.uminho.pt)

**Version:** 2 **Date:** 19 March 2009

**Author's response to reviews:** see over
Extra figure: Immunohistochemistry results of two ASC cases positive for PDGFRA and VEGFR-2, where primary antibodies were replaced by a universal negative control antibody (CEA, rabbit anti-human).